Author's response to reviews

Title: Supplier-induced demand for psychiatric admissions in Northern New England

Authors:

Bradley V Watts (bradley.v.watts@dartmouth.edu)
Brian Shiner (brian.r.shiner@dartmouth.edu)
Gunnar Klauss (gklauss@wfubmc.edu)
William B Weeks (william.b.weeks@dartmouth.edu)

Version: 5 Date: 13 August 2011

Author's response to reviews:

Dear Dr. Alam,

Thank you for accepting our paper to BMC psychiatry (MS: 1634878393498515). All changes have been made as requested below.

I apologize for the delay; I was out on vacation for a few days this week.

Thank you,

Brian Shiner MD, MPH

Authors: Bradley V Watts, Brian Shiner, Gunnar Klauss and William B Weeks
Title: Supplier-induced demand for psychiatric admissions in Northern New England
Journal: BMC Psychiatry
MS ID: 1634878393498515

Dear Prof Shiner,

We are still waiting for you to provide a revised submission to BMC Psychiatry. In order to minimise delay we have contacted all available authors.

We may have edited the manuscript for you, please ensure that all your revisions are made to the most recent version of your manuscript, which can be accessed at the following link:
http://www.biomedcentral.com/imedia/1634878393498515_article.pdf

Figures can be accessed here:
http://www.biomedcentral.com/media/man/1634878393498515/msd/FIG/1
http://www.biomedcentral.com/media/man/1634878393498515/msd/FIG/2

Please ensure that all of the formatting changes we requested have been made,
and that you have proof-read your manuscript for errors before you resubmit it.

All revised manuscripts submitted to BMC Psychiatry are regarded as being ready for publication. After publication, you will not be able to make any corrections to errors present in this version. Please check your manuscript carefully for any typing or content errors, as these will be included in the published version.

Please resubmit your manuscript here via the online submission system:

http://www.biomedcentral.com/author/manuscript/details/view.do?manuscriptId=1634878393498515

Formatting changes to be made:

----------------------------

Major revisions (we require the author to make these changes)

Tables: tables must be created using a Tables tool, and should not be linked files, figures, or generated with tabbed text. Tables must be divided into rows and columns. Please ensure that text alignment in columns is clear and consistent. We recommend the MS Word auto-format template: simple 1. Please do not include vertical lines in your tables.

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Abbreviations: Please format these in to a paragraph. Please make sure all abbreviations are spelt out and occur in alphabetical order.

Figures: It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. Our online figure guide contains full details for preparing files for submission and can be viewed here: http://www.biomedcentral.com/info/ifora/figures

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

Thank you

----------------------------

Please return your revisions by 13 August 2011

To submit your revised manuscript at any time:

1. Go to 'my BioMed Central' (http://www.biomedcentral.com/my/) and log on with your email address and password. Then click on 'My manuscripts'. You will reach a page giving details of all the manuscripts you have submitted to journals published by BioMed Central.
2. Click on the 'Submit revision' button next to the title of this manuscript.

3. On the 'Manuscript details' page, please update the title, abstract and author details if they have changed since the previous version.

4. If the two sources are different after resubmission, we will take the information in the manuscript to be the most recent and hence the most accurate and use it upon publication.

5. On the 'Cover letter' page, please provide a covering letter with a point-by-point description of the changes made.

6. On the 'Upload files' page, please upload the revised version of the manuscript and click 'Submit new version'. Please wait for the confirmation page to appear - this may take a few moments.

Please contact us if you have any problems or questions.

Best wishes,

The BMC Series Editorial Production Team
Tel: +44 (0) 20 3192 2014
Email: bmcseriesprepublication@biomedcentral.com
Web: http://www.biomedcentral.com/